Author’s response to reviews

Title: Efficacy of N,N'bis-(2-mercaptoethyl)isophthalamide on mercury intoxication: A randomized controlled trial

Authors:
Paul Schutzmeier (paul.schutzmeier@med.uni-muenchen.de)
Augusto Focil-Baquerizo (afocil@me.com)
Wilson Castillo-Tandazo (wilcas87@gmail.com)
Nicolas Focil (nfocil@fomatmedical.com)
Stephan Bose-O'Reilly (Stephan.BoeseOReilly@med.uni-muenchen.de)

Version: 3 Date: 26 Jan 2018

Author’s response to reviews:

Dear reviewer,

1. The conclusion in the abstract was rewritten. I gladly adopted your rephrased sentence, thank you for the suggestion.

2. The editors requested to add a hospitalization to the discussion. However, I added the possibility of a reimbursement to the discussion as well.

3. I rephrased the sentence to make it more comprehensible.

Thank you for all your help giving me the opportunity to improve this article and to learn.

Sincerely,

Paul Schutzmeier